Author’s response to reviews

Title: Preliminary application of 3.0T magnetic resonance chemical exchange saturation transfer imaging in brain metastasis of lung cancer

Authors:

Yonggui Yang (yangyonggui125@sina.cn)
Xiaobo QU (quxiaobo@xmu.edu.cn)
Yihui HUANG (huangyihui@stu.xmu.edu.cn)
Gang Guo (guogangxm@163.com)
Shaoxin DUAN (xmdsy@xmu.edu.cn)

Version: 3 Date: 02 Dec 2019

Author’s response to reviews:

Ethics approval and consent to participate

The research project was reviewed and approved by the Medical Ethics Committee of The Second Affiliated Hospital of Xiamen Medical College(Approval No: 2014004). All subjects were informed of the examination and signed an informed consent before the examination. Written informed consent was obtained from all participants.

Consent for publication

All presentations of case reports had consent for publication.

Availability of data and material

All data generated or analyzed during this study are included in this published article.

Competing interests

The authors declare that they have no competing interests.
Funding

This work was supported by a Planned Project Grant from the Science and Technology Bureau of Xiamen(Grant No. 3502Z20144052), a grant from the Natural Science Foundation of Fujian Province of China(Grant No. 2018J06018), and grant from the Fundamental Research Funds for the Central Universities (Grant No. 20720180056). The first funding body had a role in the design of the study, collection, analysis, and interpretation of data, and in writing the manuscript. The two remaining funding bodies had roles in the study design, interpretation of data, and in writing the manuscript.

Authors' contributions

YY designed the entire experimental protocol. As mentors, SD and GG guided the experimental process. Data collection and data analysis were performed by YY, XQ, YH, GY, GG, and SD. XQ is analytical data. All authors have been involved in drafting and revising the manuscript and approved the final version to be published. All authors read and approved the final manuscript.

Acknowledgments

The authors would like to thank everybody that contributed to this work.